
Operators Daily Log

DATE: 10-24-01 Wednesday

Time: Activities:

00:01  OPMPK on duty.

00:02 Darren from OPG called said that Time
weighted Realtime LBMP has not been
updating. The CSV file doesn’t have all of
hour 7, the html file has nothing in it and the
pdf does not have all of hour 18. It says that
it was last updated at 23:17 but I don’t see
that has  updated.
Since it is so close to the InfoRep jobs
taking off I will watch them and whether the
web is updated. If it is not I will contact
Fred T. and a price update might be in order.
(OPMPK)

00:25 The Inforep jobs ran successfully but did not
update the web.

00:30 Ran a price update for 10/23/01 but this did
not update the Time Weighted RealTime
Integrated LBMP pages. Called Fred Back
up and he said he would take a look into it.
Gave Darren a call back so he didn’t think
we had forgot about him also. (OPMPK)

01:06 Auddump ran successfully. (OPMPK)



01:16 The Time Weighted RealTime Integrated
LBMP pages have been updated. Darren of
OPG was given a call back notifying him of
this. (OPMPK)

01:30 Daytape was started. After about 5-10 min
the batch screen showed there was a input
output error. The batch screen asked if I
wanted to move the tape from 02A1 to
02A0. I replied yes the tape ejected and the
new drive became available for the daytape
so I switched drives and the job finished
successfully. (OPMPK)

01:34 The message SLS0361 swap to non library
2AO, 2A2, 2A3 came up on the batch
screen. I will send Tim a email notifying
him of this.

02:12 Daymerge ran successfully. (OPMPK)

03:29 Pros10 ran successfully. (OPMPK)

06:30 Rory Scally on site.

06:55 ReClosed HAM for BME operator.

07:40 IPL’D PROOF for Barry.

07:55 IPL’D QA.

08:15 LOAD FORECAST and CUSTOMER
SERVICE web pages are not updating. E-
mail sent to support.



09:30 Fred and Vinh stopped by and the problems
with the web pages not updating is due to a
space shortage on CLOUD 9. They will
increase the space. (RS)

11:27 IPL’D ONLINE system due to SCD not
updating.(RS)

11:30 SCD still not updating,  Mike Silver
notified. (RS)

13:05 MoveCopy for Silver in an attempt to fix the
above SCD problems.  (AI)

13:31 IPL’D ONLINE system, SCD not updating.
(RS)

13:34 ONLINE SYSTEN BACK UP, SCD now
working.(RS)

18:30 Shift Turnover


